Author's response to reviews

Title: Homozygosity and Risk of Childhood Death due to Invasive Bacterial Disease

Authors:

Emily J Lyons (e.lyons@imperial.ac.uk)
William Amos (w.amos@zoo.cam.ac.uk)
James A Berkley (jberkley@kilifi.kemri-wellcome.org)
Isaiah Mwangi (imwangi@kemri-nuitm.or.ke)
Mohammed Shafi (jshafi@afriaonline.co.ke)
Thomas N Williams (tom.n.williams@gmail.com)
Charles R Newton (CNewton@kilifi.kemri-wellcome.org)
Norbert Peshu (NPeshu@kilifi.kemri-wellcome.org)
Kevin Marsh (KMarsh@kilifi.kemri-wellcome.org)
J. Anthony G Scott (ascott@ikilifi.net)
Adrian V.S. Hill (avsh@gwmail.jr2.ox.ac.uk)

Version: 2 Date: 27 February 2009

Author's response to reviews:

No response required. This paper was submitted to BMC Biology, but has been shunted. In fact, we continue to feel this is such a striking result that it warrants a general audience. On the other hand we do want to get it published sooner rather than later, so would be happy for it to be considered by BMC Medical Genetics.